We are IntechOpen, the world's leading publisher of Open Access books Built by scientists, for scientists

5,900 Open access books available 145,000

180M



Our authors are among the

TOP 1% most cited scientists





WEB OF SCIENCE

Selection of our books indexed in the Book Citation Index in Web of Science™ Core Collection (BKCI)

# Interested in publishing with us? Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected. For more information visit www.intechopen.com



# Chapter

# The Potential of Sentinel-2 Satellite Images for Land-Cover/ Land-Use and Forest Biomass Estimation: A Review

Crismeire Isbaex and Ana Margarida Coelho

# Abstract

Mapping land-cover/land-use (LCLU) and estimating forest biomass using satellite images is a challenge given the diversity of sensors available and the heterogeneity of forests. Copernicus program served by the Sentinel satellites family and the Google Earth Engine (GEE) platform, both with free and open services accessible to its users, present a good approach for mapping vegetation and estimate forest biomass on a global, regional, or local scale, periodically and in a repeated way. The Sentinel-2 (S2) systematically acquires optical imagery and provides global monitoring data with high spatial resolution (10–60 m) images. Given the novelty of information on the use of S2 data, this chapter presents a review on LCLU maps and forest above-ground biomass (AGB) estimates, in addition to exploring the efficiency of using the GEE platform. The Sentinel data have great potential for studies on LCLU classification and forest biomass estimates. The GEE platform is a promising tool for executing complex workflows of satellite data processing.

Keywords: GEE, forest classifiers, accuracy, mapping

# 1. Introduction

In the last decades, remote sensing techniques have been applied in several studies of monitoring and classification of agricultural, forest, environmental, and socioeconomic resources [1–5]. The information extracted by a set of sensors can offer information on growth, vigor, dynamics, and diversity of vegetation cover [6–8]. In the LCLU classification and forest biomass estimation studies, the proper selection of the sensor is crucial, given the variation of spatial, radiometric, spectral, and temporal resolutions available [9]. For these studies, the use of Sentinel-2 images and a free processing platform lack information about the advantages and disadvantages between different landscapes, classification methods, and biomass estimation models.

In this way, this chapter is organized as follows: Section 2 present the satellites image classification and the potential of the Sentinel-2 satellites; Section 3 describes forest LCLU maps and the assessment of accuracy; Section 4 presents the GEE platform, advantages, and disadvantages; Section 5 describes the estimative of biomass via remote sensing; and Section 6 concludes with a platform performance overview and an outlook for the future.

## 2. The potential of image classification and the Sentinel-2 satellite

The recognition of different LCLU by remote sensing is a key parameter in the application and assessment of socioeconomic and environmental changes at local, regional, and global scales. The accurate and reliable LCLU mapping, represented by a thematic map, can be obtained through the satellite images classification [10]. In studies with a focus on forest resources, forest classification provides useful information in various decision-making processes for forest planning and management [11]. On a map, this classification becomes essential for the implementation of monitoring studies of natural and/or artificial forests, offering support in the assessment of forest protection, ecology, and quantification of carbon and biomass at different scales. In this way, data from several sensors that operate in the optical range of the electromagnetic spectrum enabled important advances in the methods of mapping and monitoring different vegetation covers.

Since the launch of the first terrestrial resource satellite in the 1950s, the analysis of vegetation data via remote sensing has been improved with advances in technology. Currently, images obtained, for example, by Landsat, SPOT, MODIS, AVHRR, ASTER, CBERS QuickBird, IKONOS, WorldView, RapidEye, Radar, LiDAR, ALOS PALSAR, and Sentinel, can produce thematic maps. In the vegetation classification process, the selection of satellite images depends on factors such as the objective of the study, availability of images, cost, level of diversity in the types of cultures, and extension of the study area [12]. In general, radar data are used to model the vertical structure of a forest and data from the multispectral optical sensors are the most used in the literature to model the horizontal structure of vegetation [13]. Multispectral sensors are capable of capturing vegetation characteristics, such as species composition, canopy cover, growth stage, and health of some forest stands [13]. Thus, different spatial, spectral, and temporal parameters that allow the extraction of robust, consistent, and comparable long-term data series must be taken into account due to the better cost-benefit ratio [14, 15]. In summary, some of the sensors used for monitoring vegetation were listed in Table 1. More information about the costs for satellite imagery can be found in http://www.landinfo.com. The sensors with medium and high spatial resolution differ in terms of the number of bands, temporal resolution, scale, and costs. Depending on the classification objective, the scale is a factor to consider, because, when choosing a sensor with a high spatial resolution (<5 m), the cost and complexity of the classification can increase [16]. In addition, with a larger set of data with spectral variability for the same class, the training time can affect the computational cost [17]. Thus, the spatial resolution must be considered as an important factor, because it must be adequate to the size of the object to be identified [16]. With images from S2, it was observed that the fragmented elements of the landscape decreased the accuracy of the classification by using the spatial resolution of 10 m, due to increased bias resulting from the total composition of pixels at the fragment edges [11].

In multispectral images, vegetation analysis can be obtained by reflectance information resulting from different wavelength bands [visible (VIS), near-infrared (NIR), and short-wave infrared (SWIR)]. The reflectance of the light spectra emits different signals according to the biophysical variables of the vegetation [18]. Among the wavelengths, the emissivity (equivalent to the absorption capacity in the thermal wave range) most used in the analysis of vegetation is in the near and middle regions of the infrared [19]. Various combinations of bands have been studied over decades to derive the biophysical variables of vegetation, resulting in a range of vegetation indices (VIs). In the LCLU classification, VIs are used to extract quantitative information from the contrast of intrinsic characteristics of spectral reflectance of vegetation [20]. VIs are useful to characterize the vigor of vegetation, pigments, sugar content and carbohydrates, high plant temperature levels,

Sensor group         Satellite/sensor         No of bands         Spatial resolution (day)         Scale of application         Data distribution policy (cost)         Data availability policy (cost)           Medium spatial resolution sensor         SPOT         4-5         2.5-20         26         L-R         Yes         1986           Aster         14         15-90         16         L-G         No         1999           IRS-P6-LISS III         4         23.5         24         L-G         No         2013           CEERS-4         4         5-20         52-26         L-R         No         2013           IRS-P6-LISS III         4         23.5         24         L-G         No         2014           ILandsat-8         11         15-100         16         L-G         No         2013           Sentinel-2         13         10-60         5         L-R         No         2015           resolution sensor         QuickBird         5         0.61-2.24         2.7         L         Yes         2007           WorldView         4-17         0.31-2.40         1-4         L         Yes         2007           GeoEye         4         0.46-1.65         2.6         <								
Sensor groupSatellite/sensorNo of bandsSpatial resolution (m)Temporal resolution (day)Scale of application*Data distribution policy (cost)Data availability policy (cost)Medium spatial resolution sensorSPOT4-52.5-2026L-RYes1986Aster1415-9016L-GNo1999IRS-P6-LISS III423.524L-GNo2003CBERS-445-2052-26L-RNo2014Landsat-81115-10016L-GNo2013Sentinel-21310-605L-RNo2015High spatial resolution sensorIKONOS51-41.5-3L-RYes1999QuickBird50.61-2.242.7LYes20072007QuickBird50.61-2.401-4LYes2007GeoEye40.46-1.652.6LYes2008RapidEye551-5.5L-RYes2012'L-G: local to global; L-R: local.L: local.Yes2012								
Medium spatial resolution sensor         SPOT $4-5$ $2.5-20$ $26$ L-R         Yes         1986           Aster         14         15-90         16         L-G         No         1999           IRS-P6-LISS III         4 $23.5$ $24$ L-G         No $2003$ CBERS-4         4 $5-20$ $52-26$ L-R         No $2014$ Landsat-8         11 $15-100$ 16         L-G         No $2013$ Sentinel-2         13 $10-60$ 5         L-R         No $2015$ High spatial resolution sensor         IKONOS         5 $1-4$ $1.5-3$ L-R         Yes         1999           QuickBird         5 $0.61-2.24$ $2.7$ L         Yes         2001           WorldView $4-17$ $0.31-2.40$ $1-4$ L         Yes $2008$ GeoEye         4 $0.46-1.65$ $2.6$ L         Yes $2008$ RapidEye         5         5 $1-5.5$ L-R         Yes $2012$ <th>Sensor group</th> <th>Satellite/sensor</th> <th>No of bands</th> <th>Spatial resolution (m)</th> <th>Temporal resolution (day)</th> <th>Scale of application<sup>a</sup></th> <th>Data distribution policy (cost)</th> <th>Data availability</th>	Sensor group	Satellite/sensor	No of bands	Spatial resolution (m)	Temporal resolution (day)	Scale of application <sup>a</sup>	Data distribution policy (cost)	Data availability
resolution sensor         Aster         14         15-90         16         L-G         No         1999           IRS-P6-LISS III         4         23,5         24         L-G         No         2003           CBERS-4         4         5-20         52-26         L-R         No         2014           Landsat-8         11         15-100         16         L-G         No         2013           Sentinel-2         13         10-60         5         L-R         No         2015           High spatial resolution sensor         IKONOS         5         1-4         1.5-3         L-R         Yes         1999           WorldView         4-17         0.31-2.40         1-4         L         Yes         2001           WorldView         4-17         0.31-2.40         1-4         L         Yes         2007           GeoEye         4         0.46-1.65         2.6         L         Yes         2008           RapidEye         5         5         1-5.5         L-R         Yes         2012	Medium spatial	SPOT	4–5	2.5–20	26	L-R	Yes	1986
IRS-P6-LISS III         4         23.5         24         L-G         No         2003           CBERS-4         4         5-20         52-26         L-R         No         2014           Landsat-8         11         15-100         16         L-G         No         2013           Sentinel-2         13         10-60         5         L-R         No         2015           High spatial resolution sensor         IKONOS         5         1-4         1.5-3         L-R         Yes         1999           QuickBird         5         0.61-2.24         2.7         L         Yes         2001           WorldView         4-17         0.31-2.40         1-4         L         Yes         2007           GeoEye         4         0.46-1.65         2.6         L         Yes         2008           RapidEye         5         5         1-5.5         L-R         Yes         2012	resolution sensor —	Aster	14	15–90	16	L-G	No	1999
CBERS-4         4         5-20         52-26         L-R         No         2014           Landsat-8         11         15-100         16         L-G         No         2013           Sentinel-2         13         10-60         5         L-R         No         2015           High spatial resolution sensor         IKONOS         5         1-4         1.5-3         L-R         Yes         1999           QuickBird         5         0.61-2.24         2.7         L         Yes         2001           WorldView         4-17         0.31-2.40         1-4         L         Yes         2007           GeoEye         4         0.46-1.65         2.6         L         Yes         2008           RapidEye         5         5         1-5.5         L-R         Yes         2012	_	IRS-P6-LISS III	4	23.5	24	L-G	No	2003
Landsat-8         11         15-100         16         L-G         No         2013           Sentinel-2         13         10-60         5         L-R         No         2015           High spatial resolution sensor         IKONOS         5         1-4         1.5-3         L-R         Yes         1999           QuickBird         5         0.61-2.24         2.7         L         Yes         2001           WorldView         4-17         0.31-2.40         1-4         L         Yes         2007           GeoEye         4         0.46-1.65         2.6         L         Yes         2008           RapidEye         5         5         1-5.5         L-R         Yes         2012	_	CBERS-4	4	5–20	52–26	L-R	No	2014
Sentinel-2         13         10–60         5         L-R         No         2015           High spatial resolution sensor         IKONOS         5         1–4         1.5–3         L-R         Yes         1999           VorldView         5         0.61–2.24         2.7         L         Yes         2001           WorldView         4–17         0.31–2.40         1–4         L         Yes         2007           GeoEye         4         0.46–1.65         2.6         L         Yes         2008           RapidEye         5         5         1–5.5         L-R         Yes         2012	_	Landsat-8	11	15–100	16	L-G	No	2013
High spatial resolution sensor         IKONOS         5         1-4         1.5-3         L-R         Yes         1999           QuickBird         5         0.61–2.24         2.7         L         Yes         2001           WorldView         4–17         0.31–2.40         1–4         L         Yes         2007           GeoEye         4         0.46–1.65         2.6         L         Yes         2008           RapidEye         5         5         1–5.5         L-R         Yes         2012	_	Sentinel-2	13	10–60	5	L-R	No	2015
Presolution sensor         QuickBird         5         0.61–2.24         2.7         L         Yes         2001           WorldView         4–17         0.31–2.40         1–4         L         Yes         2007           GeoEye         4         0.46–1.65         2.6         L         Yes         2008           RapidEye         5         5         1–5.5         L-R         Yes         2012           *L-G: local to global; L-R: local to regional; L: local.	High spatial resolution sensor — —	IKONOS	5	1–4	1.5–3	L-R	Yes	1999
WorldView         4–17         0.31–2.40         1–4         L         Yes         2007           GeoEye         4         0.46–1.65         2.6         L         Yes         2008           RapidEye         5         5         1–5.5         L-R         Yes         2012		QuickBird	-5	0.61–2.24	2.7	L	Yes	2001
GeoEye         4         0.46–1.65         2.6         L         Yes         2008           RapidEye         5         5         1–5.5         L-R         Yes         2012           "L-G: local to global; L-R: local to regional; L: local.         5         5         1–5.5         L-R         Yes         2012		WorldView	4–17	0.31–2.40	1–4	L	Yes	2007
RapidEye     5     5     1–5.5     L-R     Yes     2012 <sup>a</sup> L-G: local to global; L-R: local to regional; L: local.		GeoEye	4	0.46–1.65	2.6	L	Yes	2008
<sup>a</sup> L-G: local to global; L-R: local to regional; L: local.		RapidEye	5	5	1–5.5	L-R	Yes	2012
	<sup><i>a</i></sup> L-G: local to global; L-R:	local to regional; L: local.	$\left( \bigcirc\right)$				$\left( \bigcirc\right)$	

## Table 1.

Main characteristics of the remote sensors used for LCLU classification: spectral (number of bands), spatial and temporal resolution, data cost, the scale of application, and data availability.

and abiotic/biotic stress levels, among others [21]. Thus, VIs have become commonly applied because of computationally simple analysis of vegetation on global, regional, or local scales [20].

The forest classification is a complex task, and searching for information using individual bands can result in low accuracy models. The model input data can be represented by spectral bands combined with auxiliary bands (vegetation indices), the use of image transformation algorithms, such as principal component analysis (PCA), and image textures [13]. The combination between them helps to distinguish complex vegetation arrangements, improving thematic area estimates [13]. The vegetation indices are mathematical functions that combine two or more spectral bands [22]. The use of vegetation indices is widely discussed in the literature because they improve the classification accuracy. The VIs can reduce atmospheric interference and better distinguish the vegetation characteristics such as plant vigor, water content, sensitivity to lignin, and forest above-ground biomass (alive or dead) [23, 24]. The normalized difference vegetation index (NDVI) has been used for many years in classification studies, due to its high correlation with photosynthetically active vegetation and sensitivity to discriminate between vegetation, nonvegetation, wet, and dry areas [5, 21, 23–25].

The texture data are also important in classification studies, as it contains information of groups of pixels with similar intensity properties describing the distribution and spatial arrangement of repetitions of tones. It has been used in the quantification of the variability of pixels in a neighborhood [26]. The texture information explores information such as crown-internal shadows, size and crown shape, in coarser scales [18]. Several studies point out that the integration of texture measures increase the classification accuracy by improving the separability between classes through reducing the spectral confusion effects between spectrally similar classes [10, 27]. The incorporation of texture metrics can improve classification accuracy by up to 10 to 15% [18]. Textural information can be derived from various methods that use multiple functions of image bands in different window sizes [21]. The methods can be categorized into four main groups being based on structure, statistics, model, and transformation. As an example, we can mention the local statistics (that describe the moments of a neighborhood of individual pixels in a region of the image) and the measures of the gray level co-occurrence matrix (GLCM) (statistics set that characterize the distance and the angular relationships between pixels) [28]. In the LCLU classification, second-order GLCM texture measures have been the most used textural characteristics for characterizing the relative frequencies between the brightness values of two pixels connected by a spatial relationship [29]. However, studies that incorporate the combination of vegetation indices with image texture measurements may contribute more to improvements in classification accuracy than when they are used separately [18].

There are different classification approaches; a class can be represented based on classifiers pixel by pixel [30], per object [30], and per sub-pixel [31]. In the pixel-by-pixel classification, only spectral information of each pixel is used separately, to represent a certain class [30]. In object classification, the image segmentation is performed based on a group of pixels with similar properties, in which the algorithm examines the texture and the spectral response of the object as a basic unit, instead of individualized pixels [19]. The sub-pixel mode consists of identifying within a pixel the proportion of each type of LCLU, modifying a classification in a resolution with clear limits, without assigning mixed pixels to a dominant class [31].

In the LCLU mapping, the right choice of a classification method is essential to obtain good accuracy of representing the thematic map in an image [9]. Traditionally, the classification methods used can be supervised or unsupervised classification [32]. In particular, in supervised classification, it can be applied

parametric (*e.g.*, linear and multiple) and nonparametric [*e.g.*, the nearest neighbor K (K-NN), artificial neural network (RNA), random forest (RF), decision trees (DT), and support vector machines (SVM)] models [13, 33]. In the parametric models, the results are restricted, for example, to unimodal data, by assuming a relationship between the dependent and independent variables with an explicit model structure [13]. However, the nonparametric supervised classification has been the most used method in classification studies based on remote sensing data due to the more robust approach [33]. Thus, the supervised classification based on machine learning achieves better overall accuracy, due to the ability to deal with heterogeneous images, such as forest scenery [34–36]. Based on the training samples defined by the user, the algorithm searches for pixels belonging to each class [30]. The classifiers learn from the database and are efficient in identifying the nonlinearity of the data, noise in the samples, and can use less computation time [34].

In special, the Sentinel-2A (S2) launched in 2015, result from the Earth observation mission developed by the European Space Agency (ESA) as part of the Copernicus program, has been designed to support global monitoring for environment and security (GMES) [37]. The Sentinel-2 wide swath high-resolution Multi-Spectral Instrument system (MSI) aims to obtain information on the monitoring and management of terrestrial surface and provides continuity of the SPOT and Landsat missions [37]. With the wide coverage (swath width of 290 km) and minimum 5-day global revisit time (with twin satellites in orbit), the sensor becomes an extremely useful monitoring product for studies such as LCLU changes and environmental impacts [23]. The main level 2A output allows the use of orthoimage with corrected reflectance at the bottom-of-atmosphere (BOA) [38]. The MSI from optical sensor has 13 spectral bands, with three spatial resolutions, four bands with 10 m, including visible and near-infrared (NIR), six bands with 20 m, including four red-edge bands, two in the SWIR region, and three bands with 60 m (**Figure 1**) [39, 40].

The multispectral S2 data have several advantages over monitoring satellites available. In particular, Landsat and Sentinel-2 data are now frequently used in monitoring and management studies, to meet the demand for attributes scale data with easy access [16]. The S2 has red-edge spectral bands of longer wavelength range, essential in vegetation analysis [41]. In addition, the spatial resolution of this satellite can be



Figure 1. Multispectral bands of the Sentinel-2 in 10, 20, and 60 m spatial resolution and their wavelengths.

compared to commercial SPOT or RapidEye systems, surpassing the spatial resolution of the MODIS 250 m [41]. The Sentinel-2A satellite imagery can be acquired for free at the Sentinel Hub (https://scihub.copernicus.eu/). For the image processing, the ESA developed the Sentinel Application Platform (SNAP) software. It is a tool available to analyze satellite information for free, specialized for the Sentinel series, has a performance comparable to that attained with other software's (*e.g.*, QGIS or ENVI) [42], and has a discussion forum for consultations [43]. For these reasons, it seems to be justified to explore the potential of Sentinal-2 data for studies in forest monitoring [24] such as forest succession [2] and wildfire control [44], forest classification [11], forest management [45], and biomass estimation [46].

## 3. Maps of forest LCLU and accuracy assessment

In image classification, information on the spectral resolution of the sensors is also important to provide data on vegetation reflectance [25]. This is possible because each species emits an intrinsic electromagnetic wave according to chemical contents and morphology such as leaf structure, water content, pigments, carbohydrate and aromatic content, and proteins, among other factors [21]. In a light spectrum, remote vegetation detection is based on the ultraviolet, visible near, and medium infrared regions [47]. In the species, identification by remote sensing data requires attention, because there is spectral variability within and between healthy tree species, which can lead to an incorrect classification and hinder statistical assessments [48]. The spectrum can change, for example, with the differences in illumination, shadow effects, and during the seasons and growth periods of trees [48].

Generally, in the LCLU classification, the forest classes are considered more heterogeneous than the water surface, the exposed soil, urban and agricultural classes. The complexity of a forest canopy comes from a surface full of lighting and shading fluctuations [49]. The variability between forest classes can directly change the spectral response reflected in the images influenced by several parameters such as age, season, defoliation, the density (*e.g.*, number of trees and basal area), canopy cover, and understory [49, 50]. In remote sensing, several studies show that the gain in differentiation between species increases with the wavelength, showing a greater correction with the SWIR region [11, 46, 51]. As an example, we can mention the spectral differentiation between conifer needles and broad (flat) leaves. In general, the biggest difference between them is in transmittance and reflectance in the infrared region [51]. In relation to broad leaves, conifer needles have less transmittance at all wavelengths (greater absorption) and low reflectance in the SWIR region [11, 16, 37, 42, 43]. This effect is due to the greater sensitivity to water absorption which in conifers is deeper [51], explained by the difference in anatomical structure, biochemical composition, and thickness of the leaves, for example [11, 23, 39]. According to some studies, obtaining accurate thematic maps of the Mediterranean ecosystems multispectral bands of the Sentinel-2 in 10, 20, and 60 m spatial resolution and their wavelengths is a challenge, due to its complexity derived from the variability of tree density per unit area and similar spectral behavior [52, 53]. In the *montado* system, where there is a higher occurrence of holm oak and cork oak, in pure and mixed stands, with high spatial variability in tree densities, the effect of mixed pixels can decrease the accuracy of a classification depending on the spatial scale and resolution [54–56]. When using Landsat-8 images, the montado class can be classified as an agricultural class and vice versa, because some *montado* areas with 10 to 30% tree density can be masked by the high reflectance of the bare soil [57]. On the other hand, the *montado* areas with a tree density above 50% can be confused with the olive grove classes due to spectral proximity [57]. When using

Sentinel-2 images for canopy estimation of the *montado* system, it was observed that the use of red-edge bands, integrated in seven vegetation indices, was essential and sensitive in the vegetation properties evaluation during the summer, improving the tree crown coverage estimates [56]. The classification of seven forest species in Germany also achieved improvements in accuracy by including the SWIR bands (B11 and B12), one on the red border (B5) and two on the visible border (B2 and B4), using Sentinel-2 images and the RF algorithm [49]. Thus, one of the ways to capture the spectral confusion that can interfere with the results of a thematic map is through the evaluation of the classification model's accuracy.

In forest classification studies, the assessment of accuracy is necessary to validate a classified image because the estimates or forecasts can present errors and uncertainties [58]. The most common way to express the accuracy map is by comparing the final product area percentage with the reference data of an image. This statement is derived from one of the most used approaches to define the accuracy of the classified data, expressed in the form of the confusion matrix and/or contingency matrix [36]. Through this analysis, it is possible to generate information on overall accuracy, user's and producer's accuracy, and Kappa coefficient (k) [24, 59, 60]. The confusion matrix is a cross-tabulation, which correlates the LCLU classes in rows and columns, with reference or test data (usually represented by columns) being compared with classified or training data (usually represented by lines) [58]. In the matrix, the diagonal values indicate the number of pixels in the agreement between two class sets [61].

The overall accuracy of the thematic maps is defined by the division of number of the pixels classified correctly by the total number of samples. However, with the advances in remote sensing technology, using a computational classification in the landscape complexity, the classification standardization must be analyzed with caution. The overall accuracy is not always representative of the individual class's accuracy, that is, the high overall accuracy map does not guarantee high accuracy for the individual classes [62]. When assessing the accuracy of classification, the overall accuracy assumes a minimum value and, in each class, a value with comparable precision [63]. Therefore, to obtain more detailed accuracy information, the individual class accuracy of LCLU can also be derived from the confusion matrix. Through matrix analysis, the user's and producer's accuracy points out the errors of omission and commission, respectively. In the producer's accuracy, the user has the opportunity to evaluate the number of the pixels correctly classified in the interest class. The omitted pixels that have not been classified outside the interest class are called omission errors [58]. On the other hand, the user's accuracy indicates the percentage of pixels in a classified image that really represents the class on the ground, certifying the classification, in this case measuring commission errors [58, 64, 65]. The Kappa coefficient is used to describe the possibilities of casual agreement between predicted values and field data [66]. The K values varied from 0 to 1, where the closer to 1, the better the perfect agreement between the ground truth and the classified image [66].

In the analysis of thematic mapping, it is vital that accuracy is the best possible for each particular case, avoiding interpretation errors [67]. However, there is still no record of a preestablished minimum limit in terms of accuracy, because the reliability of a map may vary depending on the study application [65, 68, 69]. Anderson et al. [70] mentioned that an accuracy of 85% in LCLU classification maps is frequently accepted. When using MODIS images with a spatial resolution of 1 km in classification studies, Thomlinson et al. [28] defined a minimum value of 85% for overall accuracy and 70% for the classes, due to the difference in errors between spatial and thematic accuracy. The United Nations Framework Convention on Climate Change (UNFCCC) does not provide any limit to the accuracy of the data for the construction of forest reference levels [65]. Despite attempts, defining this value as a goal may be inappropriate and not represent reality in front of the challenges of the distinction of heterogeneous classes at different spatial resolutions and scales [69]. Furthermore, a classification study may be more interested in analyzing the accuracy of a particular class and disregarding others [71].

In fact, there are many studies with different classifiers that can obtain results with good accuracy [71]. Given the complexity of image classification, there are a variety of classification methods using classifying algorithms, with different approaches to achieve mapping accuracy [36]. With an emphasis on supervised classification, the machine learning algorithms exploited image data by finding hidden relationships between various input variables (parameters) with output variables (classification results) [72]. Because it is computer programming, training samples are essential to identify different spectral areas to represent a class in the image. It is based on previous samples or experiences that machine learning has the ability to generalize, that is, when trained they are able to produce solutions to an unknown data set [72]. The combination of different classification techniques has been investigated with support vector machines (SVM) [73], random forest (RF) [74], and classification and regression tree (CART) [75]. Despite the use of several methods to evaluate the performance of the algorithms, evaluating the classification accuracy is the most common one [9]. This assessment is necessary because the

Algorithms	Overall accuracy (%)	Example of use	Reference	
Artificial neural network (ANN)	65.7	Mapping of the forest vertical structure in Gong-ju, Korea.	[67]	
_	85.0ª	Discriminating urban forest types in Xuzhou, East China.	[60]	
Convolutional neural network (CNN)	90.4	Forest vegetation types in Jilin Province, China.	[77]	
_	97.7	Classification forest in Semarang, Central Java, Indonesia	[78]	
Random forest (RF)	90.9 93.2 <sup>b</sup>	Forest classification in ecosystems in Germany and South Africa.	[79]	
	88.9	Mapping of 11 forest classes in the Belgian Ardenne ecoregion, Ardenne, Belgium.	[11]	
Support vector machine (SVM)	80.0	Mapping of the invasive species American bramble ( <i>Rubus cuneifolius</i> ) in KwaZulu-Natal province of South Africa.	[80]	
RF SVM	84.2 81.8	Mapping of the crop, including high and low density forest classes in the foothill of Himalaya.	[81]	
RF K-NN SVM	94.44 95.29 94.13	Classification of six types of land use, including forest cover in the Red River Delta in Vietnam.	[82]	
_	80.0 74.3 80.3	Classification land use, including forest cover in the Dak Nong province, Vietnam.	[69]	

#### Table 2.

The overall accuracy of LCLU and forest classification with Sentinel data.

classifier's performance is affected, not only by the classifier's own limitation but also by the image quality, sample size, and computational resource [9, 76]. **Table 2** shows the overall accuracy obtained by different algorithms in some recent studies with forest classification, using Sentinel-2 images. These results aim to provide information on the current conditions of vegetation cover. In addition, it shows that the relevance of accuracy depends on the objective of the study, without being based on standard accuracy [69]. However, it was observed that classifiers have an important role in supervised classification. When using improved classification methods, such as the use of convolutional neural networks, the overall accuracy is generally higher in relation to the other methods that presented medium or fluctuating precision [36].

The studies based on forest species classification involving Sentinel-2 images are still recent [23, 33, 83–86]. The classification with S2 images of forest tree species in southwestern France obtained an overall accuracy above 90%, for plantations such as aspen (*Populus tremula*) and red oak (*Quercus rubra*). However, the species that presented more spectral confusion with an accuracy of 81 and 74% were the black pine (Pinus nigra) and Douglas fir (Pseudotsuga menziesii), respectively [87]. In tests with S2 images and the RF for forest type mapping in the Mediterranean, Italy, using four vegetation indices (NDVI, SRI, RENDVI, and ARI1) in three phenological periods (winter, spring, and summer), Puletti et al. [88] reported that the forest categories (pure coniferous forests, broadleaf forests, and mixed forests), had an overall accuracy of 86.2% and a Kappa coefficient of 0.86. The user's and producer's accuracy were above 83% for all the classes. In a study, by Duan et al. [89], they obtained overall, producer's and user's accuracy of 92.3, 92.3, and 92.2%, respectively, when mapping the distribution of urban forests in China, using eight bands and three vegetation indices (NDVI, NDWI, and NDBI) with S2 images, random forest algorithms, and the GEE platform. Thus, the spatial distribution of species and their number of trees per hectare can be understood by the classification, becoming essential in studies with different types of landscapes. When making decisions about forest resources, this information is very important, because each specific study offers support to the formulation of local and global public policies, as well as in forest planning and management up to biomass estimates.

# 4. Google Earth Engine (GEE) platform: advantages and disadvantages in the LCLU classification

In GEE, there is a computing platform that has a cloud infrastructure designed by Google, from the launch of the public data catalog of the Landsat 2008 series images [89]. The platform allows to manage, analyze, and store large volumes of geospatial historical data on a planetary scale, which can be applied for several scientific studies [89]. In the GEE, it is possible to have access to data catalog of the Landsat, MODIS, and Sentinel satellites [90]. In addition, the platform offers social, demographic, climatic, and digital elevation models and allowing the interaction remote sensing data with algorithms in synergy with the field data, using Java-Script or Python code [90]. Another cloud resource available in GEE is the Fusion Table, which offers support for tabular data, keyword-based mechanisms, and text data, among others [91].

Although the GEE platform is a free access tool, the LCLU mapping in large extensions and with high spatial resolution can be challenging for several reasons. The inclusion of large volumes of data from a complex and heterogeneous landscape requires a large computational load and processing time [92]. However, freely accessible data can be used in studies with limited funding. For example, the use of Sentinel-2 data is more recommended than the Landsat images, because it provides a slightly higher spatial and spectral resolution [5]. The classification methods can also be applied on the platform according to the image type, segmentation method, classification algorithm, training sample sets, input resources, target classes, and accuracy assessment [36]. All classification procedures can be developed on the cloud computing platform, without downloading remote sensing data, processing on desktops or other software, simplifying information extraction [86]. Therefore, the GEE platform allows the use of high-performance tools for processing a large data set. Thus, GEE is an important resource for big data management and scientific development, because it lowers barriers between the global scientific community and allows the same opportunities to share and replicate geospatial analysis [89].

With millions of servers worldwide, the platform allows the combination of different algorithms and data with free availability for noncommercial use [93]. In the image processing phase, the GEE platform can synchronize all S2 data, can easily clear cloudy pixels [86], and perform a continuous workflow of complex remote sensing [94]. According to the objective of the study, it is possible to choose the specific period and create image mosaics, with the best cloudless pixel for a specific region, solve terrain effects problems, and identify any changes in LCLU in the world through classified images [95]. Also, it can use several vegetation indices at the same time when image classification is performed [96].

Although it is of easy access, users are generally not familiar with the clientserver programming model. The GEE libraries offer a more familiar programming environment, but the user must have some basic knowledge of the programming language [95]. This requires much effort from the end user to be implemented [94]. Fortunately, there are learning platforms and discussion forums on GEE on the Internet, which help to solve most doubts and programming errors. However, difficulties have been encountered in studies of LCLU classification. Difficulties in validating the classification model carried out in the GEE were found by Zurqani et al. [96], as there was little availability of high-resolution aerial images in the world, to serve as a reference. The difficulties in image preprocessing were also reported, due to difficulties in the acquisition of parameters with the atmospheric correction [89], and limited availability of other algorithms, not allowing improvements in the classification accuracy [97].

The GEE memory defines a threshold limit to the size of the matrices, which constrains the training of the classification algorithms, with a large number of training samples and evaluation of the input bands [98]. The limited processing capacity can cause errors in complex computational analyses as in large spatial areas [95]. On the other hand, GEE processing can be affected by the very different preprocessing criteria, which makes analysis and comparison with other sensors difficult and makes GEE unsuitable for some types of processing [98, 99].

The approach to programming through the cloud platform is becoming increasingly common for large-scale computing and in multidisciplinary studies. In this way, the offer of high-resolution satellite images in the GEE can solve the problems of detecting global changes in LCLU, environmental monitoring, and help in the quantitative and qualitative identification of forest cover [24]. In particular, the Sentinel 1 (radar data) and Sentinel 2 data, have great potential for future studies of classification and estimates of above-ground biomass, through GEE. Until the present moment, few studies have tested the cloud platform capabilities with Sentinel data. This gap opens the opportunity to generate research and development in monitoring forest cover [100]. In future missions, the Sentinel data will be essential for studies to validate forest AGB estimates [100]. Free access to both resources can generate valuable information on forest cover, located in low-, medium-, and highincome countries.

## 5. Biomass estimation

Above-ground biomass (AGB) of forests plays a key role in the global carbon cycle, maintaining the climate and as bioenergy reservoirs [101]. In a global context, forest AGB or phytomass is generally defined by the quantification of stems, branches, and leaves. The estimation of the amount of the carbon mass or energy potential per unit area is also frequent [102]. In global discussions on climate change, the estimation of forest AGB has been one of the agenda items in public and private decision-making, integrating into sustainable development projects on a local to global scale [91].

In forest biomass estimates, data can be acquired using direct (destructive) methods based on harvesting the all tree and indirect (nondestructive) based on field inventory and synergy between remote sensing data [103]. The direct methods are considered to have the best accuracy, due to the dry weight determination of the parts of the tree [101]. However, when it is intended to estimate biomass, this method is time-consuming, costly and laborious, and inadequate for using in large geographic areas [13, 42]. In addition, the application of destructive methods causes disturbances in the fauna and flora with alterations in the microclimate and habitat [104]. Thus, one of the alternatives to reduce these impacts is through the use of remote sensing technology [105]. In addition to being a nondestructive method, it is based on forest inventory data, allowing the fitting of models in synergy with small-, medium-, and large-scale satellite images [91, 106].

The thematic mapping based on a forest classification is fundamental as input parameters for the area and biomass estimation [107]. It is through the accuracy of a thematic map that the user can assess the consistency of the overall reliability of the map data and accuracy measures for LCLU classes [107]. In biomass estimates by remote sensing, the type of sensor, different spatial and temporal resolutions, scale, field data, errors, and uncertainty are factors that hinder the statistical evaluation of the final product on the map. In the biomass estimate, several studies indicate errors that can range from 5 to 30% [13]. The forest planning and management decision scale can also influence the accuracy of forest biomass analysis. According to Lu et al. [13], it is recommended that in forest research, accuracy reaches values greater than 90% for a regional scale and 80% for a national or global scale.

Regardless of the estimation method used, it is fundamentally a thorough evaluation of the reference data and map data [108]. The agreement presented by the error matrix may not be equivalent to the product of the map and the reality, which impacts the biomass estimates [109]. Therefore, the classification validation analysis becomes essential because, depending on the classification methodology in a forest ecosystem, it is possible to observe errors and uncertainties that affected the classifier's performance [36]. In studies of the representation of the heterogeneity of a forest, the erroneous choice of spatial resolution can cause the addition of redundant data and can increase the noise of statistical models, without adding important information about the stands [110]. Even when an AGB forest map achieves high accuracy with the integration of multisource remote sensing data, some limitations and uncertainty can impact the results between LCLU and remote sensing data [42]. One of the limitations is found in the uncertainties about ground measurements with GPS, which can contain geolocation errors, difficult to eliminate in the image analysis processing [42]. In addition, some uncertainties in the prediction model can be included with the time gap between field data and remote sensing data [42, 111]. In this way, any gain in accuracy in biomass maps comes from advances in technology. The diversity of image processing software, greater processing capacity, and computer storage allowed the synergistic use of cloud platforms and machine learning use to deal with big data problems.

Revised studies for the Mediterranean region [106] found that when the estimates are based on passive sensor (optical data) are less accurate ( $R^2 \approx 0.70$ ) than

those carried by active sensors ( $\mathbb{R}^2 \ge 0.80$ ). One of the factors can be related to the dimension of images. While the optical sensors are based mainly on two-dimensional view—2D, creating estimates with the top canopy layer, the active sensors, such as synthetic aperture radar (SAR) and LiDAR, reach the third dimension of the forest with the evaluation of the arboreal and understory, presenting a better data correlation with the biomass [106]. Among some studies, it was found that the spatial resolution has a key role in the accuracy of the biomass. The spatial resolution of Landsat can achieve a low accuracy [112], while the SPOT satellites [113], GeoEye [114], and Quickbird [55] can achieve medium to high accuracy, WorldView [115] and spatial resolution of LiDAR can achieve high accuracy [116]. Overall, the more accurate the modeling, the greater the approximation to the observed values [117]. Accurate models with optical data can be achieved using the high spatial resolution (<10 m), where the pixel size approximates the size of the study object [118]. In particular, using Sentinel-2 data, it has been reported medium accuracy in local and regional scale studies [119, 120]. Thus, the production of AGB maps, with Sentinel-2 data, has a great potential to expand to forest management and monitoring decisions on a regional scale.

In quantifying the biomass stock with Sentinel data, the cross-use of sensor data combined with forest inventory data and algorithms were fundamental for gains in

Satellite	Location	Forest settings	Model performance		Reference	
data		_	R <sup>2</sup>	RMSE (Mg. ha <sup>-1</sup> )		
S2	Evros prefecture, Rhodopes mountain range, Greece	Mediterranean forest	0.63	63.11 <sup>ª</sup>	[122]	
S2	Parsa National Park, Nepal	Central-southern part of Nepal, subtropical climate	0.81	25.32	[124]	
S2	Parque Nacional Yok Don, Vietnam	Tropical monsoon climate	0.81	36.67	[119]	
S2	Parsa National Park, Nepal	Central-southern part of Nepal, subtropical climate	0.99	4.51	[125]	
S2	Hunan Province, southern China	Subtropical monsoon climate	0.58	65.03ª	[123]	
S1 and S2	Island province of Palawan, Philippines	Southern coast of Honda Bay within the administrative jurisdiction of Puerto Princesa City, tropical climate	0.75	33.81	[43]	
S1 and S2	Ecoregion of Changbai Mountains mixed forests and eastern mountainous region of Jilin Province in northeast China	Monsoon- influenced humid continental climate	0.97	33.29	[76]	

#### Table 3.

Studies on forest biomass estimation using algorithms RF in different ecological settings.

the accuracy of the estimates. A study by Castillo et al. [43] showed improvements in the accuracy of biomass models with Sentinel-1 (S1) and S2, which were comparable to the image accuracy of current commercial sensors. In biomass estimates, precision can also differ depending on the order of importance of spectral bands and textures, according to the input characteristics in the model [121]. When analyzing Landsat-8 satellite images with the inclusion of S2 images, the importance of SWIR region reflectance to improve the estimation of forest parameters was observed [122]. In study using the GEE platform, it was identified that the greatest contribution of the variables in the estimates of forest AGB were those composed by the SWIR and red-edge bands [119]. In addition, Hu et al. [123] reported that the red-edge band (B5) of S2 was the best in the prediction with good performance of the RF algorithm on the GEE platform. In particular, the RF algorithm showed good accuracy in AGB estimates in subtropical deciduous forests, as it was robust with the nonlinearity of the data [124]. Due to the large reported of the random forest algorithm in biomass estimation studies, Table 3 summarizes the results of some recent studies that used Sentinel (S1 and S2) data to estimate the forest AGB in different scenarios. It was observed that the machine learning algorithm was used in different climatic conditions and in time series with considerable accuracy.

Although advances in image resolution and use of radar data have been increasingly available, the storing of a large amount of data and with a complex data structure is one of the major challenges of research with the geospatial data [126]. With the advances in technology, cloud computing has helped researchers to solve big data problems, which reduces the costs of accessing software and maintaining hardware [93]. However, to date, studies with cloud platforms such as GEE for forest biomass estimates have been little explored. For forest AGB, cloud computing holds promise in solving problems related to big data [126]. The platform's advantages are related to the data storage and analysis process of an intensive nature and based on complex point structures, such as LIDAR data [91]. The GEE platform is being used as rasterized data management in parallel with other software such as packages in the R [125] and cloud computing applications such as Fusion Tables and Google Cloud Platform [91]. In the near future, it is hoped that it will be possible to integrate data such as high-precision LiDAR with a collection of optical images for mapping global biomass [91].

In remote sensing, the promising prospects for biomass estimates can take new directions thanks to the ESA's Earth Explorer mission Biomass. The mission scheduled for this decade aims to provide global maps of biomass and carbon stored in the world's forests [126]. As a novelty, the Biomass will have the first P-band synthetic aperture radar capable of carrying out the precise mapping of biomass estimates [126]. In addition, it will have an experimental tomograph to provide 3D views of the forests [126]. In this way, we hope that in the future, it will be possible to carry out studies with crossing Sentinel 2 data with Biomass to compose thematic maps of LCLU changes and biomass stock with greater accuracy.

## 6. Conclusions and outlook

The development of LCLU maps for biomass estimates with Sentinel images is still recent, but promising. In studies of LCLU classification, the overall accuracy and producer's and user's accuracy should be the highest possible for better support of biomass prediction models. For the refinement of the classification map, the accuracy can be improved with the combination of radar and optical data from Sentinel 1 and 2, respectively, as well as incorporating models and algorithms, vegetation indices, textures, biophysical variables, and forest inventory data, among others. However, it is noteworthy that the accuracy of the biomass prediction models obtained by remote sensing still depends on the precision of the field-based measurements.

The GEE platform allows the use of free data, combined with the remote data and machine learning for building maps, in addition, to assess the use of land occupation on a large scale. The inclusion of samples collected in the field in the Google Fusion Table and multitemporal data, such as high-resolution images can improve the classification results.

Based on the limitations of the GEE platform, we hope that the new updates of the platform can solve problems such as memory space and the inclusion of other ranking algorithms. The improvements make the platform more accessible and attract new users to assess changes in LCLU and analysis of vegetation cover monitoring. In future works, the LCLU classification approach will be based on plots using the pixel method connected to the GEE and field validation, which facilitates the detection and understanding of the dynamics of forest areas in terms of volumetric and gravimetric production of biomass over time.

To our knowledge, studies that address the accuracy of a forest classification combined with biomass estimates, using Sentinel images on cloud platforms, have not yet been reported in the literature. Available studies make separate approaches to forest classification and biomass estimates. With Sentinel imagery, these two themes generally use field data and other thematic maps to develop the models. This limitation shows the opportunity to develop pioneering research on forest classification and biomass estimates with Sentinel images on cloud platforms.

# Acknowledgements

The work was supported by Programa Operativo de Cooperação Transfronteiriço Espanha-Portugal (POCTEP); project CILIFO – Centro Ibérico para la Investigación y Lucha contra Incendios Forestales and by FCT, Portugal, Fundação para a Ciência e Tecnologia, through IDMEC, under LAETA, project UIDB/05183/2020.

# IntechOpen

# **Author details**

Crismeire Isbaex<sup>\*</sup> and Ana Margarida Coelho Department of Rural Engineering, School of Sciences and Technology, Institute of Research and Advanced Information (IIFA), University of Évora, Évora, Portugal

\*Address all correspondence to: cisbaex@uevora.pt

## IntechOpen

© 2020 The Author(s). Licensee IntechOpen. Distributed under the terms of the Creative Commons Attribution - NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/), which permits use, distribution and reproduction for non-commercial purposes, provided the original is properly cited.

# References

[1] Askar, Nuthammachot N, Phairuang W, Wicaksono P, Sayektiningsih T. Estimating aboveground biomass on private forest using sentinel-2 imagery. Journal of Sensors. 2018;**2018**:1-11. DOI: 10.1155/2018/6745629

[2] Szostak M, Hawryło P, Piela D. Using of Sentinel-2 images for automation of the forest succession detection.
European Journal of Remote Sensing. 2018;51(1):142-149. DOI: 10.1080/22797254.2017.1412272

[3] Caiserman A, Dumas D, Bennafla K, Faour G, Amiraslani F. Application of remotely sensed imagery and socioeconomic surveys to map crop choices in the Bekaa Valley (Lebanon). Agriculture. 2019;**9**(3):1-19. DOI: 10.3390/agriculture9030057

[4] Carranza C, Benninga H, van der Velde R, van der Ploeg M. Monitoring agricultural field trafficability using Sentinel-1. Agricultural Water Management. 2019;**224**:1-12. DOI: 10.1016/j.agwat.2019.105698

[5] Ganivet E, Bloomberg M. Towards rapid assessments of tree species diversity and structure in fragmented tropical forests: A review of perspectives offered by remotely-sensed and field-based data. Forest Ecology and Management. 2019;**432**:40-53. DOI: 10.1016/j.foreco.2018.09.003

[6] Mulla DJ. Twenty five years of remote sensing in precision agriculture: Key advances and remaining knowledge gaps. Biosystems Engineering. 2013;**114**(4):358-371. DOI: 10.1016/j. biosystemseng.2012.08.009

[7] Battude M, Al Bitar A, Morin D, Cros J, Huc M, Marais Sicre C, et al. Estimating maize biomass and yield over large areas using high spatial and temporal resolution Sentinel-2 like remote sensing data. Remote Sensing of Environment. 2016;**184**:668-681. DOI: 10.1016/j.rse.2016.07.030

[8] Ali A, Ullah S, Bushra S, Ahmad N, Ali A. Quantifying forest carbon stocks by integrating satellite images and forest inventory data. Austrian Journal of Forest Science. 2018;2:93-117

[9] Lu D, Weng Q. A survey of image classification methods and techniques for improving classification performance. International Journal of Remote Sensing. 2007;**28**(5):823-870. DOI: 10.1080/01431160600746456

[10] Li M, Zang S, Zhang B, Li S, Wu C. A review of remote sensing image classification techniques: The role of spatio-contextual information. European Journal of Remote Sensing. 2014;**47**(1):389-411. DOI: 10.5721/ EuJRS20144723

[11] Bolyn C, Michez A, Gaucher P, Lejeune P, Bonnet S. Forest mapping and species composition using supervised per pixel classification of Sentinel-2 imagery. Biotechnology, Agronomy, Society and Environment. 2018;**22**(3):172-187. DOI: 10.25518/1780-4507.16524

[12] Zheng B, Myint SW, Thenkabail PS, Aggarwal RM. A support vector machine to identify irrigated crop types using time-series Landsat NDVI data. International Journal of Applied Earth Observation and Geoinformation. 2015;**34**(1):103-112. DOI: 10.1016/j.jag.2014.07.002

[13] Lu D, Chen Q, Wang G, Liu L, Li G, Moran E. A survey of remote sensing-based aboveground biomass estimation methods in forest ecosystems. International Journal of Digital Earth. 2016;**9**(1):63-105. DOI: 10.1080/17538947.2014.990526

[14] Alilou H, Moghaddam Nia A, Keshtkar H, Han D, Bray M. A cost-effective and efficient framework to determine water quality monitoring network locations. The Science of the Total Environment. 2018;**624**:283-293. DOI: 10.1016/j.scitotenv.2017.12.121

[15] Turner W, Rondinini C, Pettorelli N, Mora B, Leidner AK, Szantoi Z, et al. Free and open-access satellite data are key to biodiversity conservation. Biological Conservation. 2015;**182**:173-176. DOI: 10.1016/j.biocon.2014.11.048

[16] Chen Y, Guerschman JP, Cheng Z, Guo L. Remote sensing for vegetation monitoring in carbon capture storage regions: A review. Applied Energy.
2019;240:312-326. DOI: 10.1016/j. apenergy.2019.02.027

[17] Spiekermann R, Brandt M, Samimi C. Woody vegetation and land cover changes in the Sahel of Mali (1967-2011). International Journal of Applied Earth Observation and Geoinformation. 2015;**34**(1):113-121. DOI: 10.1016/j.jag.2014.08.007

[18] Fassnacht FE, Latifi H, Stereńczak K, Modzelewska A, Lefsky M, Waser LT, et al. Review of studies on tree species classification from remotely sensed data. Remote Sensing of Environment. 2016;**186**: 64-87. DOI: 10.1016/j.rse.2016.08.013

[19] Bernardi H, Dzedzej M, Carvalho L, Acerbi JF. Classificação digital do uso do solo comparando os métodos "pixel a pixel" e orientada ao objeto em imagem QuickBird. In: XIII Simpósio Brasileiro de Sensoriamento Remoto. Vol. 2001. 2007. pp. 5595-5602

[20] Frampton WJ, Dash J, Watmough G, Milton EJ. Evaluating the capabilities of Sentinel-2 for quantitative estimation of biophysical variables in vegetation. ISPRS Journal of Photogrammetry and Remote Sensing. 2013;**82**:83-92. DOI: 10.1016/j.isprsjprs.2013.04.007

[21] Xue J, Su B. Significant remote sensing vegetation indices: A review

of developments and applications. Journal of Sensors. 2017;**2017**:1-17. DOI: 10.1155/2017/1353691

[22] Ku NW, Popescu SC. A comparison of multiple methods for mapping local-scale mesquite tree aboveground biomass with remotely sensed data. Biomass and Bioenergy. 2019;**122**:270-279. DOI: 10.1016/j. biombioe.2019.01.045

[23] Haywood A, Stone C, Jones S. The potential of Sentinel satellites for large area aboveground forest biomass mapping. In: International Geoscience and Remote Sensing Symposium (IGARSS). 2018. pp. 9030-9033. DOI: 10.1109/IGARSS.2018.8517597

[24] Koskinen J, Leinonen U, Vollrath A, Ortmann A, Lindquist E, D'Annunzio R, et al. Participatory mapping of forest plantations with Open Foris and Google Earth Engine. ISPRS Journal of Photogrammetry and Remote Sensing. 2019;**148**:63-74. DOI: 10.1016/j.isprsjprs.2018.12.011

[25] Solymosi K, Kövér G, Romvári R.
The progression of vegetation indices: A short overview. Acta Agraria
Kaposváriensis. 2019;23(1):75-90. DOI: 10.31914/aak.2264

[26] Jensen JR. Remote Sensing of the Environment: An Earth Resource Perspective 2/e. New Delhi: Pearson Education India; 2009. p. 613

[27] Chen Q, Gong P. Automatic variogram parameter extraction for textural classification of the panchromatic IKONOS imagery. IEEE Transactions on Geoscience and Remote Sensing. 2004;**42**(5):1106-1115. DOI: 10.1109/TGRS.2004.825591

[28] Thomlinson JR, Bolstad PV, Cohen WB. Coordinating methodologies for scaling landcover classifications from site-specific to global: Steps toward validating global

map products. Remote Sensing of Environment. 1999;**70**(1):16-28. DOI: 10.1016/S0034-4257(99)00055-3

[29] Mananze S, Pôças I, Cunha M. Mapping and assessing the dynamics of shifting agricultural landscapes using Google Earth Engine cloud computing, a case study in Mozambique. Remote Sensing. 2020;**12**(8):1-23. DOI: 10.3390/ RS12081279

[30] Furtado LFA, Francisco CN, Almedida CM. Análise de imagem baseada em objeto para classificação das fisionomias da vegetação em imagens de alta resolução espacial. Geociências. 2013;**32**(3):441-451

[31] Wu K, Du Q, Wang Y,
Yang Y. Supervised sub-pixel mapping for change detection from remotely sensed images with different resolutions. Remote Sensing.
2017;9(3):1-17. DOI: 10.3390/rs9030284

[32] Lillesand TM, Kiefer RW. RemoteSensing and Image Interpretation.New York: John Wiley & Sons, inc.;2000. p. 736

[33] Thanh Noi P, Kappas M. Comparison of random forest, k-nearest neighbor, and support vector machine classifiers for land cover classification using Sentinel-2 imagery. Sensors (Basel, Switzerland). 2018;**18**(1):1-20. DOI: 10.3390/s18010018

[34] Belgiu M, Drăgu L. Random forest in remote sensing: A review of applications and future directions. ISPRS Journal of Photogrammetry and Remote Sensing. 2016;**114**:24-31. DOI: 10.1016/j.isprsjprs.2016.01.011

[35] Fu B, Wang Y, Campbell A, Li Y, Zhang B, Yin S, et al. Comparison of object-based and pixel-based random forest algorithm for wetland vegetation mapping using high spatial resolution GF-1 and SAR data. Ecological Indicators. 2017;**73**:105-117. DOI: 10.1016/j.ecolind.2016.09.029 [36] Ma L, Li M, Ma X, Cheng L, Du P, Liu Y. A review of supervised objectbased land-cover image classification. ISPRS Journal of Photogrammetry and Remote Sensing. 2017;**130**:277-293. DOI: 10.1016/j.isprsjprs.2017.06.001

[37] Roteta E, Bastarrika A, Padilla M, Storm T, Chuvieco E. Development of a Sentinel-2 burned area algorithm: Generation of a small fire database for sub-Saharan Africa. Remote Sensing of Environment. 2019;**222**:1-17. DOI: 10.1016/j.rse.2018.12.011

[38] ESA. Sentinel-2 User Handbook.Vol. 2. European Space Agency ESAStandard Document. Paris, France;2015. p. 64. DOI: 10.1021/ie51400a018

[39] Chang J, Shoshany M. Mediterranean shrublands biomass estimation using Sentinel-1 and Sentinel-2. In: IEEE International Geoscience and Remote Sensing Symposium (IGARSS). Beijing, China; 10-15th July 2016. pp. 5300-5303. DOI: 10.1109/IGARSS.2016.7730380

[40] ESA. Spatial Resolution [Internet].2020. Available from: https://sentinel.esa.int/web/sentinel/user-guides/sentinel-2-msi/resolutions/spatial

[41] Hawryło P, Bednarz B, Wężyk P, Szostak M. Estimating defoliation of Scots pine stands using machine learning methods and vegetation indices of Sentinel-2. European Journal of Remote Sensing. 2018;**51**(1):194-204. DOI: 10.1080/22797254.2017.1417745

[42] Zhang T, Su J, Liu C, Chen WH, Liu H, Liu G. Band selection in Sentinel-2 satellite for agriculture applications. In: 2017 23rd International Conference on Automation and Computing (ICAC). IEEE; 2017. pp. 1-6

[43] Castillo JAA, Apan AA, Maraseni TN, Salmo SG. Estimation and mapping of above-ground biomass of mangrove forests and their replacement land uses in the Philippines using Sentinel imagery. ISPRS Journal of Photogrammetry and Remote Sensing. 2017;**134**:70-85. DOI: 10.1016/j.isprsjprs.2017.10.016

[44] Fernandes PM, Guiomar N, Rossa CG. Analysing eucalypt expansion in Portugal as a fire-regime modifier. The Science of the Total Environment. 2019;**666**:79-88. DOI: 10.1016/j.scitotenv.2019.02.237

[45] Barton I, Király G, Czimber K, Hollaus M, Pfeifer N. Treefall gap mapping using Sentinel-2 images. Forests. 2017;**8**(426):1-27. DOI: 10.3390/f8110426

[46] Thomas N, Simard M, Castañeda-Moya E, Byrd K, Windham-Myers L, Bevington A, et al. High-resolution mapping of biomass and distribution of marsh and forested wetlands in southeastern coastal Louisiana. International Journal of Applied Earth Observation and Geoinformation. 2019;**80**:257-267. DOI: 10.1016/j. jag.2019.03.013

[47] Randive PU, Deshmukh RR, Janse PV, Kayte JN. Study of detecting plant diseases using non-destructive methods: A review. International Journal of Emerging Trends & Technology in Computer Science. 2018;7(1):66-71

[48] Waser LT, Küchler M, Jütte K, Stampfer T. Evaluating the potential of worldview-2 data to classify tree species and different levels of ash mortality. Remote Sensing. 2014;**6**:4515-4545. DOI: 10.3390/rs6054515

[49] Immitzer M, Vuolo F, Atzberger C. First experience with Sentinel-2 data for crop and tree species classifications in Central Europe. Remote Sensing. 2016;**8**(3):1-27. DOI: 10.3390/rs8030166

[50] Landry S, St-Laurent MH, Nelson PR, Pelletier G, Villard MA. Canopy cover estimation from Landsat images: Understory impact on topof-canopy reflectance in a northern hardwood forest. Canadian Journal of Remote Sensing. 2018;**44**(5):435-446. DOI: 10.1080/07038992.2018.1533399

[51] Rautiainen M, Lukeš P, Homolová L, Hovi A, Pisek J, Mõttus M. Spectral properties of coniferous forests: A review of in situ and laboratory measurements. Remote Sensing. 2018;**10**(2):1-28. DOI: 10.3390/rs10020207

[52] Van Doorn AM, Pinto CT.
Differences in land cover interpretation in landscapes rich in cover gradients: Reflections based on the montado of South Portugal. Agroforestry Systems.
2007;70(2):169-183. DOI: 10.1007/ s10457-007-9055-8

[53] Allen H, Simonson W, Parham E, Santos EDBE, Hotham P. Satellite remote sensing of land cover change in a mixed agro-silvo-pastoral landscape in the Alentejo, Portugal. International Journal of Remote Sensing. 2018;**39**(14):4663-4683. DOI: 10.1080/01431161.2018.1440095

[54] Sousa AMO, Gonçalves AC, Mesquita P, Marques da Silva JR. Biomass estimation with high resolution satellite images: A case study of Quercus rotundifolia. ISPRS Journal of Photogrammetry and Remote Sensing. 2015;**101**:69-79. DOI: 10.1016/j. isprsjprs.2014.12.004

[55] Macedo FL, Sousa AMO, Gonçalves AC, Marques da Silva JR, Mesquita PA, RAF R. Above-ground biomass estimation for Quercus rotundifolia using vegetation indices derived from high spatial resolution satellite images. European Journal of Remote Sensing. 2018;**51**(1):932-944. DOI: 10.1080/22797254.2018.1521250

[56] Godinho S, Guiomar N, Gil A. Estimating tree canopy cover percentage in a Mediterranean silvopastoral systems using Sentinel-2A imagery and the stochastic gradient boosting algorithm. International Journal of Remote Sensing. 2018;**39**(14):4640-4662. DOI: 10.1080/01431161.2017.1399480

[57] Godinho S, Guiomar N, Gil A. Using a stochastic gradient boosting algorithm to analyse the effectiveness of Landsat 8 data for montado land cover mapping: Application in southern Portugal. International Journal of Applied Earth Observation and Geoinformation. 2016;**49**:151-162. DOI: 10.1016/j. jag.2016.02.008

[58] Nguyen HTT, Doan TM, Tomppo E, McRoberts RE. Land use/land cover mapping using multitemporal Sentinel-2 imagery and four classification methods—A case study from Dak Nong, Vietnam. Remote Sensing. 2020;**12**(1367):1-27. DOI: 10.3390/rs12091367

[59] Xiong J, Thenkabail PS, Tilton JC, Gumma MK, Teluguntla P, Oliphant A, et al. Nominal 30-m cropland extent map of continental Africa by integrating pixel-based and object-based algorithms using Sentinel-2 and Landsat-8 data on Google Earth Engine. Remote Sensing. 2017;**9**(10):1-27. DOI: 10.3390/rs9101065

[60] Lee YS, Lee S, Jung HS. Mapping forest vertical structure in Gong-Ju, Korea using Sentinel-2 satellite images and artificial neural networks. Applied Sciences. 2020;**10**(5):1-18. DOI: 10.3390/ app10051666

[61] Banko G. A review of assessing the accuracy of and of methods including remote sensing data in forest inventory. Laxenburg, Austria: International Institute for Applied Systems Analysis; 1998. IR-98-081. pp. 1-42

[62] GFOI. Integrating Remote-Sensing and Ground-Based Observations for Estimation of Emissions and Removals of Greenhouse Gases in Forests. Geneva, Switzerland: Group on Earth; 2013. p. 164

[63] Foody GM. Status of land cover classification accuracy assessment.
Remote Sensing of Environment.
2002;80(1):185-201. DOI: 10.1016/ S0034-4257(01)00295-4 [64] Story M, Congalton RG. Remote sensing brief accuracy assessment: A user's perspective. Photogrammetric Engineering and Remote Sensing. 1986;**52**(3):397-399

[65] FAO. Map accuracy assessment and area estimation: A practical guide. In: National Forest Monitoring Assessment Working Paper No. 46/E. Rome, Italy: Food and Agriculture Organization of the United Nations; 2016. p. 69. Available from: http://www.fao.org/3/ai5601e.pdf

[66] Liu C, Frazier P, Kumar L. Comparative assessment of the measures of thematic classification accuracy. Remote Sensing of Environment. 2007;**107**(4):606-616. DOI: 10.1016/j.rse.2006.10.010

[67] Stehman SV. Selecting and interpreting measures of thematic classification accuracy. Remote Sensing of Environment. 1997;**62**:77-89. DOI: 10.1016/S0034-4257(97)00083-7

[68] Foody GM. Explaining the unsuitability of the kappa coefficient in the assessment and comparison of the accuracy of thematic maps obtained by image classification. Remote Sensing of Environment. 2020;**239**:111630: 1-11. DOI: 10.1016/j.rse.2019.111630

[69] Lark RM. Components of accuracy of maps with special reference to discriminant analysis on remote sensor data. International Journal of Remote Sensing. 1995;**16**:1461-1480. DOI: 10.1080/01431169508954488

[70] Anderson JR, Hardy EE, Roach JT, Witmer RE. A Land Use and Land Cover Classification System for Use with Remote Sensor Data. Washington, DC, USA: U.S. Government Publishing Office; 1976

[71] Sothe C, de Almeida CM, Liesenberg V, Schimalski MB. Evaluating Sentinel-2 and Landsat-8 data to map sucessional forest stages in a subtropical forest in southern Brazil. Remote Sensing. 2017;**9**(8):1-23. DOI: 10.3390/ rs9080838

[72] Farda NM. Multi-temporal land use mapping of coastal wetlands area using machine learning in Google Earth Engine. IOP Conference Series: Earth and Environmental Science. 2017;**98**(1):1-23. DOI: 10.1088/1755-1315/98/1/012042

[73] Mountrakis G, Im J, Ogole C. Support vector machines in remote sensing: A review. ISPRS Journal of Photogrammetry and Remote Sensing. 2011;**66**(3):247-259. DOI: 10.1016/j. isprsjprs.2010.11.001

[74] Pal M. Random forest classifier for remote sensing classification. International Journal of Remote Sensing. 2005;**26**(1):217-222. DOI: 10.1080/01431160412331269698

[75] Breiman L, Friedman JH, Olshen RA, Stone CJ. Classification and Regression Trees. New York, USA: Chapman & Hall; 1984. p. 358. DOI: 10.1002/cyto.990080516

[76] Chen L, Wang Y, Ren C, Zhang B, Wang Z. Assessment of multiwavelength SAR and multispectral instrument data for forest aboveground biomass mapping using random forest kriging. Forest Ecology and Management. 2019;447:12-25. DOI: 10.1016/j.foreco.2019.05.057

[77] Zhou X, Li L, Chen L, Liu Y, Cui Y, Zhang Y, et al. Discriminating urban forest types from Sentinel-2A image data through linear spectral mixture analysis: A case study of Xuzhou, East China. Forests. 2019;**10**(6):1-19. DOI: 10.3390/f10060478

[78] Liu H, Gu L, Ren R, He F. Classification of forest vegetation types in Jilin Province, China based on deep learning and multi-temporal Sentinel-2 data. In: Proc. SPIE 11127, Earth Observing Systems XXIV, 1112725. pp. 1-12. DOI: 10.1117/12.2527392

[79] Miranda E, Mutiara AB, Ernastuti, Wibowo WC. Forest classification method based on convolutional neural networks and Sentinel-2 satellite imagery. International Journal of Fuzzy Logic and Intelligent Systems. 2019;**19**(4):272-282. DOI: 10.5391/ IJFIS.2019.19.4.272

[80] Heckel K, Urban M, Schratz P, Mahecha MD, Schmullius C. Predicting forest cover in distinct ecosystems: The potential of multi-source Sentinel-1 and -2 data fusion. Remote Sensing. 2020;**12**(2):1-22. DOI: 10.3390/ rs12020302

[81] Rajah P, Odindi J, Mutanga O, Kiala Z. The utility of Sentinel-2 vegetation indices (VIs) and Sentinel-1 synthetic aperture radar (SAR) for invasive alien species detection and mapping. Natural Conservation. 2019;**35**:41-61. DOI: 10.3897/ natureconservation.35.29588

[82] Saini R, Ghosh SK. Crop classification on single date Sentinel-2 imagery using random forest and support vector machine. In: ISPRS
- International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences. Vol. XLII-5. 2018. pp. 683-688. DOI: 10.5194/ isprs-archives-xlii-5-683-2018

[83] Chatziantoniou A, Petropoulos GP, Psomiadis E. Co-orbital Sentinel 1 and 2 for LULC mapping with emphasis on wetlands in a Mediterranean setting based on machine learning. Remote Sensing. 2017;9(12):1-20. DOI: 10.3390/rs9121259

[84] Fragoso-Campón L, Quirós E, Mora J, Gutiérrez JA, Durán-Barroso P. Accuracy enhancement for land cover classification using LIDAR and multitemporal Sentinel 2 images in a forested watershed. Proceedings. 2018;**2**(1280):1-4. DOI: 10.3390/ proceedings2201280

[85] Tieng T, Sharma S, Mackenzie RA, Venkattappa M, Sasaki NK, Collin A. Mapping mangrove forest cover using Landsat-8 imagery, Sentinel-2, very high resolution images and Google Earth Engine algorithm for entire Cambodia. IOP Conference Series: Earth and Environmental Science. 2019;**266**(1):1-12. DOI: 10.1088/1755-1315/266/1/012010

[86] Li H, Jia M, Zhang R, Ren Y, Wen X. Incorporating the plant phenological trajectory into mangrove species mapping with dense time series Sentinel-2 imagery and the Google Earth Engine platform. Remote Sensing. 2019;**11**(21):1-16. DOI: 10.3390/rs11212479

[87] Karasiak N, Sheeren D, Fauvel M, Willm J, Dejoux JF, Monteil C. Mapping tree species of forests in Southwest France using Sentinel-2 image time series. In: 2017 9th International Workshop on the Analysis of Multitemporal Remote Sensing Images, MultiTemp 2017. 2017. pp. 1-4. DOI: 10.1109/Multi-Temp.2017.8035215

[88] Puletti N, Chianucci F, Castaldi C. Use of Sentinel-2 for forest classification in Mediterranean environments. Annals of Silvicultural Research. 2017;**42**(1):1-7. DOI: 10.12899/ASR-1463

[89] Duan Q, Tan M, Guo Y, Wang X, Xin L. Understanding the spatial distribution of urban forests in China using Sentinel-2 images with Google Earth Engine. Forests. 2019;**10**(729):1-15. DOI: 10.3390/f10090729

[90] Mutanga O, Kumar L. Google Earth Engine applications. Remote Sensing. 2019;**11**(5):1-4. DOI: 10.3390/rs11050591

[91] Oliphant AJ, Thenkabail PS, Teluguntla P, Xiong J, Gumma MK, Congalton RG, et al. Mapping cropland extent of southeast and Northeast Asia using multi-year time-series Landsat 30-m data using a random forest classifier on the Google Earth Engine cloud. International Journal of Applied Earth Observation and Geoinformation. 2019;**81**:110-124. DOI: 10.1016/j. jag.2018.11.014

[92] Maxwell AE, Strager MP, Warner TA, Ramezan CA, Morgan AN, Pauley CE. Large-area, high spatial resolution land cover mapping using random forests, GEOBIA, and NAIP orthophotography: Findings and recommendations. Remote Sensing. 2019;**11**(12):1-27. DOI: 10.3390/ rs11121409

[93] Dong J, Xiao X, Menarguez MA, Zhang G, Qin Y, Thau D, et al. Mapping paddy rice planting area in northeastern Asia with Landsat 8 images, phenology-based algorithm and Google Earth Engine. Remote Sensing of Environment. 2016;**185**:142-154. DOI: 10.1016/j.rse.2016.02.016

[94] Shelestov A, Lavreniuk M, Kussul N, Novikov A, Skakun S. Exploring Google Earth Engine platform for big data processing: Classification of multi-temporal satellite imagery for crop mapping. Frontiers in Earth Science. 2017;5:1-10. DOI: 10.3389/ feart.2017.00017

[95] Gorelick N, Hancher M, Dixon M, Ilyushchenko S, Thau D, Moore R. Google Earth Engine: Planetary-scale geospatial analysis for everyone. Remote Sensing of Environment. 2017;**202**:18-27. DOI: 10.1016/j.rse.2017.06.031

[96] Zurqani HA, Post CJ, Mikhailova EA, Schlautman MA, Sharp JL. Geospatial analysis of land use change in the Savannah River Basin using Google Earth Engine. International Journal of Applied Earth Observation and Geoinformation. 2018;**69**:175-185. DOI: 10.1016/j.jag.2017.12.006

[97] Camara G, Assis LF, Ribeiro G, Ferreira KR, Llapa E, Vinhas L, et al. Big earth observation data analytics: Matching requirements to system architectures. In: Proceedings of the 5th ACM SIGSPATIAL International Workshop on Analytics for Big Geospatial Data, BigSpatial. Vol. 3. 2016. pp. 1-6. DOI: 10.1145/3006386.3006393

[98] Carrasco L, O'Neil AW, Daniel Morton R, Rowland CS. Evaluating combinations of temporally aggregated Sentinel-1, Sentinel-2 and Landsat 8 for land cover mapping with Google Earth Engine. Remote Sensing. 2019;11:1-21. DOI: 10.3390/rs11030288

[99] Coluzzi R, Imbrenda V, Lanfredi M, Simoniello T. A first assessment of the Sentinel-2 level 1-C cloud mask product to support informed surface analyses. Remote Sensing of Environment. 2018;**217**:426-443. DOI: 10.1016/j. rse.2018.08.009

[100] Ghosh SM, Behera MD. Aboveground biomass estimation using multi-sensor data synergy and machine learning algorithms in a dense tropical forest. Applied Geography. 2018;**96**:29-40. DOI: 10.1016/j.apgeog.2018.05.011

[101] Sun X, Li G, Wang M, Fan Z. Analyzing the uncertainty of estimating forest aboveground biomass using optical imagery and spaceborne LiDAR. Remote Sensing. 2019;**11**(6):1-6. DOI: 10.3390/rs11060722

[102] Pertille CT, Nicoletti MF, Topanotti LR, Stepka TF. Biomass quantification of *Pinus taeda* L. from remote optical sensor data. Advances in Forestry Science. 2019;**6**(2):603-610. DOI: 10.34062/afs.v6i2.7086

[103] Kushwaha SPS, Nandy S, Gupta M. Growing stock and woody biomass assessment in Asola-Bhatti Wildlife Sanctuary, Delhi, India. Environmental Monitoring and Assessment. 2014;**186**(9):5911-5920. DOI: 10.1007/ s10661-014-3828-0

[104] Vashum KT, Jayakumar J. Methods to estimate above-ground biomass and carbon stock in natural forests - A review. Journal of Ecosystem and Ecography. 2012;**02**(04):1-8. DOI: 10.4172/2157-7625.1000116

[105] Timothy D, Onisimo M, Riyad I. Quantifying aboveground biomass in African environments: A review of the trade-offs between sensor estimation accuracy and costs. Tropical Ecology. 2016;**57**(3):393-405

[106] Galidaki G, Zianis D, Gitas I, Radoglou K, Karathanassi V, Tsakiri-Strati M, et al. Vegetation biomass estimation with remote sensing: Focus on forest and other wooded land over the Mediterranean ecosystem. International Journal of Remote Sensing. 2017;**38**(7):1940-1966. DOI: 10.1080/01431161.2016.1266113

[107] Duncanson L, Armston J, Disney M, Avitabile V, Barbier N, Calders K, et al. The importance of consistent global forest aboveground biomass product validation. Surveys in Geophysics. 2019;**40**(4):979-999. DOI: 10.1007/ s10712-019-09538-8

[108] Ali I, Greifeneder F, Stamenkovic J, Neumann M, Notarnicola C. Review of machine learning approaches for biomass and soil moisture retrievals from remote sensing data. Remote Sensing. 2015;7(12):16398-16421. DOI: 10.3390/rs71215841

[109] Natarajan K, Latva-Käyrä P, Zyadin A, Pelkonen P. New methodological approach for biomass resource assessment in India using GIS application and land use/land cover (LULC) maps. Renewable and Sustainable Energy Reviews. 2016;**63**:256-268. DOI: 10.1016/j.rser.2016.05.070

[110] Bajwa SG, Bajcsy P, Groves P, Tian LF. Hyperspectral image data mining for band selection in agricultural applications. Transactions of ASAE. 2004;**47**(3):895-908

[111] Song C, Chen JM, Hwang T, Gonsamo A, Croft H, Zhang Q, et al.

Ecological characterization of vegetation using multisensor remote sensing in the solar reflective spectrum. In: Thenkabail PS, editor. Land Resources Monitoring, Modeling, and Mapping with Remote Sensing. Remote Sensing Handbook. London, UK: Taylor and Francis; 2015;2:533-575. DOI: 10.1201/ b19322

[112] Sánchez-RuizS, Moreno-MartínezÁ, Izquierdo-Verdiguier E, Chiesi M, Maselli F, Gilabert MA. Growing stock volume from multi-temporal landsat imagery through Google Earth Engine. International Journal of Applied Earth Observation and Geoinformation. 2019;**83**:1-10. 101913. DOI: 10.1016/j. jag.2019.101913

[113] Wang X, Wang S, Dai L.
Estimating and mapping forest biomass in Northeast China using joint forest resources inventory and remote sensing data. Journal of Forestry Research.
2018;29(3):797-811. DOI: 10.1007/ s11676-017-0504-6

[114] Qazi WA, Baig S, Gilani H, Waqar MM, Dhakal A, Ammar A. Comparison of forest aboveground biomass estimates from passive and active remote sensing sensors over Kayar Khola watershed, Chitwan district, Nepal. Journal of Applied Remote Sensing. 2017;**11**(2):1-17. DOI: 10.1117/1.jrs.11.026038

[115] Rasel SMM, Chang HC, Ralph TJ, Saintilan N, Diti IJ. Application of feature selection methods and machine learning algorithms for saltmarsh biomass estimation using Worldview-2 imagery. Geocarto International. 2019:1-25. DOI: 10.1080/10106049.2019.1624988

[116] Leite RV, Hummel C, Pires RDP, Silva CA, Pedro C, Soares B, et al. Estimating stem volume in eucalyptus plantations using airborne LiDAR: A comparison of area- and individual tree-based approaches. Remote Sensing. 2020;**12**(9):1-26. DOI: 10.3390/ rs12091513

[117] Sanquetta CR. Métodos de determinação de biomassa florestal. In: Sanquetta CR et al, editors. As florestas e o carbono. Curitiba: Universidade Federal do Paraná; 2002. pp.119-140

[118] Roth KL, Roberts DA, Dennison PE, Peterson SH, Alonzo M. The impact of spatial resolution on the classification of plant species and functional types within imaging spectrometer data. Remote Sensing of Environment. 2015;**171**:45-57. DOI: 10.1016/j.rse.2015.10.004

[119] Dang ATN, Nandy S, Srinet R, Luong NV, Ghosh S, Senthil KA. Forest aboveground biomass estimation using machine learning regression algorithm in Yok Don National Park, Vietnam. Ecological Informatics. 2019;**50**:24-32. DOI: 10.1016/j.ecoinf.2018.12.010

[120] Liu Y, Gong W, Xing Y, Hu X, Gong J. Estimation of the forest stand mean height and aboveground biomass in Northeast China using SAR Sentinel-1B, multispectral Sentinel-2A, and DEM imagery. ISPRS Journal of Photogrammetry and Remote Sensing. 2019;**151**:277-289. DOI: 10.1016/j. isprsjprs.2019.03.016

[121] Astola H, Häme T, Sirro L, Molinier M, Kilpi J. Comparison of Sentinel-2 and Landsat 8 imagery for forest variable prediction in boreal region. Remote Sensing of Environment. 2019;**223**(2018):257-273. DOI: 10.1016/j.rse.2019.01.019

[122] Chrysafis I, Mallinis G, Siachalou S, Patias P. Assessing the relationships between growing stock volume and Sentinel-2 imagery in a Mediterranean forest ecosystem. Remote Sensing Letters. 2017;8(6):508-517. DOI: 10.1080/2150704X.2017.1295479

[123] Hu Y, Xu X, Wu F, Sun Z, Xia H, Meng Q, et al. Estimating forest stock volume in Hunan Province, China, by integrating in situ plot data, Sentinel-2 images, and linear and machine learning regression models. Remote Sensing. 2020;**12**(1):1-23. DOI: 10.3390/rs12010186

[124] Pandit S, Tsuyuki S, Dube T. Estimating above-ground biomass in sub-tropical buffer zone community forests, Nepal, using Sentinel 2 data. Remote Sensing. 2018;**10**(4):1-18. DOI: 10.3390/rs10040601

[125] Pandit S, Tsuyuki S, Dube T. Exploring the inclusion of Sentinel-2 MSI texture metrics in aboveground biomass estimation in the community forest of Nepal. Geocarto International. 2019;**34**:1-18. DOI: 10.1080/10106049.2019.1588390

[126] EAS. Biomass [Internet]. 2020. Available from: https:// earth.esa.int/web/guest/missions/ esa-future-missions/biomass

